Author's response to reviews

**Title:** Differential endothelial cell gene expression by African Americans versus Caucasian Americans: A possible contribution to health disparity in vascular disease and cancer

**Authors:**

Peng Wei (Peng.Wei@uth.tmc.edu)
Liming Chang Milbauer (chang032@umn.edu)
Judy Enenstein (enens001@umn.edu)
Julia Nguyen (nguye088@umn.edu)
Wei Pan (weip@biostat.umn.edu)
Robert P Hebbel (hebbe001@umn.edu)

**Version:** 5 **Date:** 19 November 2010

**Author's response to reviews:** see over
19 November 2010

Editorial Office
BMC Medicine

RE: BMC Medicine, MS:4758905994073540

Dear Editors:

Our final revision of this manuscript is enclosed. The only changes to text (indicated in red) are on page 5 as follows:

old reference number [6] now appears correctly as [8]
old reference number [7] now appears correctly as [9]

In addition, a revised version of Table 3 is also submitted now. The two parts have now been combined into one table

Sincerely,

Robert P. Hebbel, M.D.
University of Minnesota Medical School